\documentclass[11pt,letterpaper]{article}
\usepackage{amssymb,array,amsmath,times}



%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%% CHANGE THE THREE VARIABLES BELOW %%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

\def\lectureno{1}
\def\dateofclass{12.03.2012}
\def\nameofscribe{Alice B. Cooper}

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

\setlength{\topmargin}{-0.2in}
\setlength{\textheight}{8.9in}   % 11.0 - 1.125 - 0.875
\setlength{\textwidth}{6.2in}    %  8.5 - 1.375 - 1.125
\setlength{\oddsidemargin}{0.0in}
\setlength{\evensidemargin}{0.0in}

\newtheorem{theorem}{Theorem}
\newtheorem{assumption}{Assumption}
\newtheorem{lemma}[theorem]{Lemma}
\newtheorem{remark}{Remark}
\newtheorem{definition}[theorem]{Definition}
\newtheorem{corollary}[theorem]{Corollary}
\newtheorem{proposition}[theorem]{Proposition}
\newenvironment{proof}{\noindent {\em Proof.}}{\medskip}
\newcommand{\ignore}[1]{}
\def\squareforqed{\hbox{\rlap{$\sqcap$}$\sqcup$}}
\def\qed{\ifmmode\squareforqed\else{\unskip\nobreak\hfill\penalty50\hskip1em\null\nobreak\hfil\squareforqed\parfillskip=0pt\finalhyphendemerits=0\endgraf}\fi}

\begin{document}

\begin{tabular}{|p{10cm}p{5cm}|}
\hline
{\em Bonn-Aachen International Center for Information Technology} & \\
Cryptography and Game Theory & \hfill Raekow, Ziegler \\
IPEC Winter 2012 &\\
& \\
Scribe(s): \nameofscribe &  \hfill Lecture \lectureno, Date: \dateofclass \\
\hline
\end{tabular}
\vspace{2cm}

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%% START WRITING THE LECTURE HERE %%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

\section{Part 1}
In this part of the lecture
\subsection{SubPart 1}
The details that we discussed were the following:
\begin{itemize}
\item a 
\item b
\item c
\end{itemize}
\subsection{SubPart 2}
More stuff is discussed here.
\section{Part 2}
The following formula holds for all $\epsilon$
\[
\epsilon > 0.
\]







\end{document}